Author's response to reviews

Title: Changes in Practice Patterns Affecting In-Hospital and Post-Discharge Survival Among Acute Coronary Syndrome Patients

Authors:

Manfred Stommel (stommel@msu.edu)
Ade Olomu (ade.olomu@ht.msu.edu)
Margaret Holmes-Rovner (mholmes@msu.edu)
William Corser (corser@msu.edu)
Joseph C Gardiner (jgardiner@epi.msu.edu)

Version: 4 Date: 19 October 2006

Author's response to reviews: see over
Thursday, October 19, 2006

Iratxe Puebla
Senior Assistant Editor
BMC-series journals

Dear Ms. Puebla,

We are pleased that your reviewers found our revisions to be adequate. As requested, we made the following formatting changes:
1. We rewrote the abstract background section to reflect the place of the current study in the current literature.
2. We added the IRB information in the Methods section - we apologize for this oversight.
3. We changed 'Conclusion' to Conclusions' in the text.
4. We added the sentence on authors' approval to the Authors' Contributions paragraph.
5. We cropped the figures to minimize white space.
Finally, we followed Reviewer 2 to rearrange Table 1 with respect to the number of vessels listed. We hope these responses meet your expectations.

Sincerely,

Manfred Stommel, PhD
Associate Professor, CON
W-149 Owen Graduate Center
Michigan State University
East Lansing, MI 48824
Phone: 517-355-5123
Fax: 517-353-9553
E-mail: stommel@msu.edu